Author’s response to reviews

Title: Infantile Hepatic Hemangiomas associated with high-output cardiac failure and pulmonary hypertension ------a case report and review of the literature

Authors:
Xintong Zhang (zhangxt29@163.com)
Weidong Ren (renwdcmu@163.com)
Guang Song (songg84@qq.com)
Yangjie Xiao (xyjcmu@163.com)
Feifei Sun (839521531@qq.com)
Nan Wang (elainenannan@qq.com)

Version: 1 Date: 04 Sep 2019

Author’s response to reviews:

Dear editor,

We sincerely thank the editor and the reviewers for giving us constructive and in-depth suggestions which have helped us. Here we submit a new version of our manuscript with the title “Infantile Hepatic Hemangiomas associated with high-output cardiac failure and pulmonary hypertension ------a case report and review of the literature”, which has been modified according to the reviewers’ suggestions. We mark all the changes in yellow in the revised manuscript.

Reviewer1: “I would edit English quite extensively before considering this paper for publication…”
Response: We have revised the manuscript to by reorganizing verbal expressions and have improve our English with the assistance from another native speaker.

Reviewer1: “I would describe pathophysiology of heart failure a bit more, underscoring the role for increased systemic venous….. as a potential mechanism for pulmonary hypertension in this disease.”
Response: We have replenished the pathophysiology of heart failure as well as the potential mechanism of pulmonary hypertension in the yellow part (PAGE5, LINE21-31) in the article.

Reviewer2: “… review the literature and expand the topic”
Response: The literature review consists of 3 tables including summary of the literature on patients presenting with IHH associated with congestive heart failure, treatments and outcomes. We also add more descriptions to the topic.( Table1 in PAGE10-11, Table 2 in PAGE12, Table 3 in PAGE13)

Once again, thank you so much for the excellent and professional suggestions of our manuscript.

Best wishes